˛�Gèt SÜľuą(ÂąÙ), 2001[ 1ą§

˛_ěÒ : 301.

ž¿p¡ : ˛ÇŢ(27-303A, 880-8796, hkang@math.snu.ac.kr) ˛˚è^ Ď ˛˚ì : Ü, @ 1:00-2:15, 24-417

„Å: ¸ ˛˚ĆtŹ ‘yÛŞAéÁ, SÜľuąY ě¡ľuą˚ ¸Áú ĘÖœ£ ăŁ, ;ê– Ń[Cő ŃSĹ [CĆ Âć /ê˚ ¸Áú –G:œ£ WÙćř.

(1) Dirichlet-to-Neumann map (2) Boundary determination and layer stripping (3) Determination of isotropic conductivity in 3D (4) Determination of isotropic conductivity in 2D, I (5) Determination of isotropic conductivity in 2D, II (6) Current state of determiantion of anisotropic conductivity

œL[ű :

[1 ] G. Uhlmann, The Dirichlet to Neumann Map and Inverse Problems, Lecture Note (unpublished) [2 ] V. Isakov, Inverse problems for PDEs, Springer, 1998

e` : |^LP (written), §˙LP (oral)